Thu May 14 16:57:10 GMI 1998
Year: 98 Doy: 134
Observer: koon
WEATHER COMMENT: Thu May 14 16:57:12 GMTI 1998
Medi um cirrostratus overcast, wi nd=5 nmph fromthe East, tenp=50 F
MKI Il COMMENT: Thu May 14 16:59:28 GMI 1998

Wwn't start this up due to clouds, will work on cable and Iimt switch
changes.

Thu May 14 17:00:22 GV 1998 CH P Startup--Initializing new tape
Thu May 14 17:05:16 GVI 1998 CH P CH P Start Patro

Thu May 14 17:05:16 GMI 1998 CH P CH P Start Patro

LONML COWMENT: Thu May 14 17:08: 04 GVIT 1998

L0O0515 was popped out of tenmp drive #0, |’m not sure but that m ght have
been as it should have done. Rempved it and installed LO0517 in tenp

drive #0.

PICS COMVENT: Thu May 14 17:35:26 GMI 1998

Pausi ng this one due to clouds.

PSPT COMVENT: Thu May 14 17:35:37 GMI 1998

I’mnot sure what the status of this instrument is, yesterday Haosheng said
| should wait for e-mail fromhimbefore starting up today, haven't
received any e-mail fromhimyet.

End Patr ol
Thu May 14 18:03: 05 GVI 1998 CH P Bi as
Thu May 14 18:04: 07 GVI 1998 CH P End Bi as
Thu May 14 18:04:15 GMTI 1998 CH P Wat er
Thu May 14 18:04:55 GMTI 1998 CH P End Water
Thu May 14 18:05: 03 GMI 1998 CH P Vel ocity
Thu May 14 18:05:36 GVI' 1998 CH P End Vel ocity

CH P COWMENT: Thu May 14 18:14:51 GVII 1998
The instrument appears to be working nornally after the hourly velocity
measur enment sequence.

Thu May 14 19:03: 08 GVI 1998 CH P Bi as

Thu May 14 19:04: 04 GVI 1998 CH P End Bi as

Thu May 14 19:04: 12 GV 1998 CH P Wat er

Thu May 14 19:04:57 GMI 1998 CH P End Water
Thu May 14 19:05: 04 GMI 1998 CH P Vel ocity

Thu May 14 19: 05: 40 GMI 1998 CH P End Vel ocity
PSPT COMMENT: Thu May 14 19:44:56 GMI 1998

Haosheng called, | opened the donme and got things ready for him
Thu May 14 20:03: 06 GVI 1998 CH P Gai n

Thu May 14 20:09: 08 GMI 1998 CH P End Gain

Thu May 14 20:09:15 GMI 1998 CH P Bi as

Thu May 14 20: 10: 07 GVI 1998 CH P End Bi as

Thu May 14 20: 10: 15 GVI 1998 CH P Wat er

Thu May 14 20:10: 55 GV 1998 CH P End Water



Thu May 14 20:11:02 GVI 1998 CH P Vel ocity

Thu May 14 20: 11: 50 GMI 1998 CH P End Vel ocity
COWENT: Thu May 14 20:33:49 GVI 1998

Ext ended t he done slot.

Thu May 14 21:01: 06 GMI 1998 CH P Bi as

Thu May 14 21:02: 03 GMI 1998 CH P End Bi as

Thu May 14 21:02: 10 GVI 1998 CH P Wat er

Thu May 14 21:03: 04 GMI 1998 CH P End Water
Thu May 14 21:03:12 GVI 1998 CH P Vel ocity

Thu May 14 21:03:46 GMI 1998 CH P End Vel ocity

PSPT COWMENT: Thu May 14 21:45:25 GMTI 1998
Ri gged up a tenporary light baffle around the dec axis limt switch to see
if ambient light is getting into the sensor and confusi ng things.

Thu May 14 22:01: 03 GMI 1998 CH P Bi as

Thu May 14 22:01: 58 GMI 1998 CH P End Bi as

Thu May 14 22:02: 05 GMI 1998 CH P Wat er

Thu May 14 22:02:49 GMI 1998 CH P End Water

Thu May 14 22:02:57 GMI 1998 CH P Vel ocity

Thu May 14 22:03:32 GV 1998 CH P End Vel ocity

PSPT COMMENT: Thu May 14 22:40: 48 GMTI 1998

Still haven't fixed the declination pointing problem 1'Il |eave the various
el ectroni cs on overni ght so Haosheng can work on it. I'Il take the done out of
conputer control after | close it, to be safe.

Thu May 14 22:46:02 GV 1998 CH P CH P End Patro

MKI'TI COMMVENT: Thu May 14 23:16: 15 GMIr 1998

| didn't have enough time to finish the dark shutter Iimt switch installation
due to other tasks. But | did figure out how the circuitry works and

exactly where to nount the new switches and nade sone progress renoving
uneeded circuitry and wiring and making the new nmounts. It should go

fairly fast and easy now, |I'Il get dark shutter hooked up and tested on
Saturday, and if the weather permits | should be able to do the ND filter

too. Right now the dark shutter isn’t functional. On the spar, both the

ND filter and the dark shutter nechani sm cabl es are unpl ugged.

COMMENT: Thu May 14 23:37:36 GMI 1998

TAPES:

*kkk k%

MKI'lTl: Too cloudy for data
PI CS: Too cloudy for data
CH P: (00581

LOAL: LO0516 in tenp drive #1

pe
Thu May 14 23:42: 02 GMI 1998
Ml T



